Author's response to reviews

Title: Genotyping and clinical characteristics of multidrug and extensively drug-resistant tuberculosis from a tertiary care tuberculosis hospital in China

Authors:

Xiaoliang Yuan (yxlyyx@126.com)
Tiantuo Zhang (zhtituli@163.com)
Kazuyoshi Kawakami (kawakami@med.tohoku.ac.jp)
Jiaxin Zhu (zhujiaxin2020@163.com)
Hongtao Li (LHT791@163.com)
Wenzheng Zheng (utkvsutk@163.com)
Guofang Deng (dgf7210@163.com)
Shaohua Tu (1525783211@qq.com)
Weiyou Liu (gzweiyou@yahoo.com.cn)

Version: 3 Date: 13 June 2013

Author's response to reviews: see over
Dear Editor:

Recently we have remodified our manuscript, entitled “Genotyping and clinical characteristics of multidrug and extensively drug-resistant tuberculosis in a tertiary care tuberculosis hospital in China” (MS: 1204211201862874), in response to the comments and suggestions from the reviewers and editor. Please note that the paper was modified with a specific focus on improving the phrasing throughout the document, and that many of the changes in revised manuscript were made to reduce wordiness and improve the flow of the paper. We would like to submit the enclosed, revised manuscript for further consideration.

We would like to thank the Editor for the assistance provided in processing this manuscript.

Best wishes,

Xiaoliang Yuan
Division of Respiratory Diseases, Department of Internal Medicine,
The 3rd Affiliated Hospital of Sun Yat-sen University,
Institute of Respiratory Diseases of Sun Yat-Sen University,
600 Tianhe Road,
Guangzhou 510630, China
Phone: 86-20-85253084.
Fax: 86-20-85252874.
E-mail: yxlyyx@126.com.